Author's response to reviews

Title: Spontaneous intraperitoneal rupture of hepatic hydatid cyst: a rare cause of ascites

Authors:

manel limeme (limeme.manel@hotmail.com)
sana yahyaoui (dr.sana2010@yahoo.fr)
mosaab ghannouchi (badreddinebouzid.bouzid@gmail.com)
abdellajid khnissi (badreddinebouzid.bouzid2013@gmail.com)
houneida zaghouani (houneida7@yahoo.fr)
habib amara (badrbzd@yahoo.fr)
dejla bakir (Amalmes10@gmail.com)
rached letaief (simsouma2009@hotmail.fr)
chakib kraiem (sim.sem2009@live.fr)

Version: 2
Date: 13 January 2014

Author's response to reviews: see over
Dear the Editor,

We would like to submit the enclosed manuscript “Spontaneous intraperitoneal rupture of hepatic hydatid cyst: a rare cause of ascites” for your consideration for publication in BMC surgery Journal. This case report concern intraperitoneal rupture of hepatic hydatid cyst revealed by ascites. Ascites is rarely seen in the course of hydatid disease, but can result from cyst rupture into the peritoneal cavity. The diagnosis of ruptured hydatid cyst should be prompt because it requires emergency intervention.

The manuscript has been approved by all authors and has never been published, or under the consideration for publication elsewhere.

In the revised version, I provide all the Acknowledgements requested.

Yours Sincerely,

Yahyaoui Sana

E-mail: dr.sana2010@yahoo.fr

Adress: Department of radiology, Farhat Hached hospital, 4000 Sousse, Tunisia.